Author’s response to reviews

Title: Value-based genomic screening: Exploring genomic screening for chronic diseases using triple value principles

Authors:

Viktor Dombrádi (dombradi.viktor@sph.unideb.hu)
Erica Pitini (ericapiti@hotmail.it)
Carla van El (cg.vanel@vumc.nl)
Anant Jani (anant.jani@phc.ox.ac.uk)
Martina Cornel (MC.Cornel@vumc.nl)
Paolo Villari (paolo.villari@uniroma1.it)
Muir Gray (muir.gray@phc.ox.ac.uk)
Klára Bíró (kbiro@med.unideb.hu)

Version: 2 Date: 30 Oct 2019

Author’s response to reviews:

Dr. James Edwards
Assistant Editor
BMC Health Services Research

Dear Dr. James Edwards,

Thank you for giving us the opportunity to publish our work in your scholarly journal.

As requested, the following changes were made:

(1) In both the abstract and the main body of text “Discussion” was renamed to “Main text”.

(2) At the end of the funding we added the following statement: “The funders had no involvement in the design of the study and collection, analysis and interpretation of data or writing the manuscript.”

(3) In the manuscript all track changes were accepted. During the final proofread minimal changes, such as correcting typos, were made.

(4) Also, one of the authors, Sir Professor Muir Gray asked to include "Oxford Centre for Triple Value Healthcare, Oxford, United Kingdom" as his affiliation. Thus, this was added next to his name.

We hope that you find these changes adequate.

We would also like to thank both reviewers for putting substantial time and effort to review our manuscript. Their feedbacks and recommendations have improved the manuscript significantly. For this we are most grateful.

Sincerely,

The authors of the manuscript